**Author's response to reviews**

**Title:** Comparison between two methods of working length determination and its effect on radiographic extent of root canal filling: a clinical study

**Authors:**
Leena L M Smadi (alsomadi@yahoo.com)

**Version:** 5  **Date:** 29 December 2005

**Author's response to reviews:**

Dear Sir:
Thank you for considering my manuscript for publication. Your comments were of great help.
In response to your comments the following changes have been made:
* Abstract: Background has been expanded.
* Material and Methods has been renamed methods
* References list has been formatted as requested
* English language has been reviewed

Sincerely yours
Dr. Smadi Leena BDS MDent SCi FDSRCS